

How Does the Seasonal Cycle Control Equatorial Atlantic Interannual Variability?

C. Prodhomme, Aurore Voldoire, E. Exarchou, A.-l. Deppenmeier, J.

García-Serrano, Virginie Guemas

► To cite this version:

C. Prodhomme, Aurore Voldoire, E. Exarchou, A.-l. Deppenmeier, J. García-Serrano, et al.. How Does the Seasonal Cycle Control Equatorial Atlantic Interannual Variability?. Geophysical Research Letters, 2019, 46 (2), pp.916-922. 10.1029/2018GL080837 . hal-04914594

HAL Id: hal-04914594 https://cnrs.hal.science/hal-04914594v1

Submitted on 30 Jan 2025 $\,$

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Copyright



Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL080837

Key Points:

- The Atlantic Niño bias weakly depends on the climatological SST, wind, and thermocline errors
- Errors in the representation of the seasonal cold tongue development affect the simulation of the Atlantic Niño
- Biases in the Bjerknes feedback climatology and variability are related to each other

Correspondence to:

C. Prodhomme, chloe.prodhomme@ub.edu

Citation:

Prodhomme, C., Voldoire, A., Exarchou, E., Deppenmeier, A.-L., García-Serrano, J., & Guemas, V. (2019). How does the seasonal cycle control equatorial Atlantic interannual variability? *Geophysical Research Letters*, *46*, 916–922. https://doi.org/10.1029/2018GL080837

Received 10 OCT 2018 Accepted 30 DEC 2018 Accepted article online 9 JAN 2019 Published online 23 JAN 2019

How Does the Seasonal Cycle Control Equatorial Atlantic Interannual Variability?

C. Prodhomme^{1,2}, A. Voldoire³, E. Exarchou², A.-L. Deppenmeier^{4,5}, J. García-Serrano^{1,2}, and V. Guemas²

¹Group of Meteorology, Universitat de Barcelona (UB), Barcelona, Spain, ²Barcelona Supercomputing Center (BSC), Barcelona, Spain, ³Centre National de Recherches Météorologiques (CNRM), Météo-France/CNRS UMR3589, Toulouse, France, ⁴Wageningen University, Meteorology and Air Quality, Wageningen, The Netherlands, ⁵Royal Netherlands Meteorological Institute (KNMI), RD Weather and Climate Models, De Bilt, The Netherlands

Abstract It has long been thought that reducing mean-state biases would lead to improvements in variability. However, so far, there is no confirmation of a relation between model mean biases and variability. While most coupled models exhibit substantial sea surface temperature (SST) biases in the Tropical Atlantic, they are still able to reproduce reasonable SST variability in the basin. We investigate the relationship between the first- and second-order moments of the SST distribution in the equatorial Atlantic using Coupled Model Intercomparison Project Phase 5 simulations. Results suggest that the ability of the coupled models to properly reproduce the interannual variability is linked to their ability to simulate a realistic seasonal cycle evolution, that is, a realistic cold tongue development and a realistic Bjerknes feedback during the beginning of summer, rather than to their ability to represent the summer SST climatology.

Plain Language Summary To study climate change and natural climate variability, the climate community uses global coupled models. However, these models are not perfect; model systematic errors or biases appear due to approximations in the equations and misrepresentation of different parametrized processes. Biases affect all timescales: the climatology, the seasonal cycle evolution, and interannual variability. In the Tropical Atlantic, almost all state-of-the-art coupled models show very large biases in sea surface temperature. At the same time, this region has a very strong seasonal cycle and a leading mode of variability at equatorial latitudes called the Atlantic Niño. In the present study, by analyzing 36 coupled models, we conclude that it is essential to properly simulate the seasonal cycle evolution to yield a realistic simulation of the Atlantic Niño.

1. Introduction

In the Tropical Atlantic (TA), the climatological sea surface temperature (SST) is strongly biased in most Coupled Model Intercomparison Project Phase 5 (CMIP5) coupled models (Figure 1, Wang et al., 2014). TA SST also displays a marked seasonal cycle, with the development of a strong cold tongue in the eastern part of the basin from May to July (Merle et al., 2013, Figure 2a). In spring, when SST starts to cool in the eastern TA due to the northward migration of the local Inter Tropical Convergence Zone, the easterlies along the equator increase. These easterlies shallow the thermocline in the east, which leads to enhanced SST cooling (Okumura et al., 2004). This ocean-atmosphere coupling is regulated by the Bjerknes feedback (Bjerknes, 1969) and is poorly represented in the CMIP5 models (Richter et al., 2014). Wind errors seem responsible for the wrong settlement of the cold tongue in summer (Richter et al., 2014; Voldoire et al., 2014), but erroneous heat fluxes might play a role in the existence of the warm SST bias throughout the year (Hourdin et al., 2015; Voldoire et al., 2014).

TA SST displays large interannual variability, which strongly affects surrounding population through fishery (Bacha et al., 2017) and modulation of the West African monsoon (Rodríguez-Fonseca et al., 2015). The TA interannual variability also influences the El Niño-Southern Oscillation (Rodríguez-Fonseca et al., 2009; Terray et al., 2015), the Indian Monsoon (Kucharski et al., 2008), and European climate (Cassou et al., 2005). The main SST mode of variability at interannual timescale is the equatorial zonal mode or Atlantic Niño, occurring with a periodicity of 2–5 years (Carton et al., 1996). The Atlantic Niño peaks during boreal summer,

©2019. American Geophysical Union. All Rights Reserved.

Geophysical Research Letters



Figure 1. (a) Observed mean sea surface temperature (SST) in JJ. (b) Multimodel (MM) mean SST in JJ. (c) Difference between MM and observed mean JJ SST. (d) Observed standard deviation of SST in JJ ($\sigma_{JJ}SST$). (e) MM mean of $\sigma_{JJ}SST$. (f) Difference between MM and observed $\Delta_{JJ}SST$. (g) Observed $\Delta_{JJ}SST$. (h) MM mean $\Delta_{JJ}SST$. (i) Difference between MM and observed $\Delta_{JJ}SST$. (b) MM mean $\Delta_{JJ}SST$. (c) Difference between MM and observed $\Delta_{JJ}SST$. (b) MM mean $\Delta_{JJ}SST$. (c) Difference between MM and observed $\Delta_{JJ}SST$. For all panels, reference data set is ERSST 1979–2014, unit ° C; black box shows the Atl3 region. For panels (a) and (c) the contour shows ERSST 1870–1900. Panels (c), (f), and (i): the areas where MM bias is smaller than the difference between ERSST and HadSST are shaded

when the cold tongue develops. The Atlantic Niño, described as the anomalous state of the cold tongue, has mainly been attributed to the Bjerknes feedback (Keenlyside & Latif, 2007; Zebiak, 1993) whereby it is strongly tied to the seasonal cycle (Burls et al., 2011).

Despite the large biases of climatological SST in the region, models are able to reproduce reasonably the Atlantic Niño and the Bjerknes feedback (Deppenmeier et al., 2016). Ding et al. (2015) and Dippe et al. (2017) show that an improved mean state of the TA can lead to better representation of interannual variability in the Kiel climate model, suggesting that error in the climatological SST could be related with errors in the interannual variability.

This study investigates how erroneous SST background, and/or erroneous SST seasonal cycle, affect the representation of interannual variability in the CMIP5 models. It aims at complementing two previous studies using the CMIP5 models in the TA: Richter et al. (2014), who investigated the relation between winds and SST biases and variability and Deppenmeier et al. (2016), whose analysis assessed the representation of the interannual variability in the Bjerknes feedback.

The paper is organized as follows: Section 2 presents the data and methods, section 3 presents the analysis of the CMIP5 data, and section 4 summarizes and discusses the results.

2. Data and Methods

2.1. CMIP5 Database

The model simulations used in this study are obtained from CMIP5 (Taylor et al., 2012). We use one member of the preindustrial control runs. For the thermocline depth, we use the 20° isotherm computed from 3-D monthly data. In the rest of the study, a color is associated to each model; the legend can be found in Figure 2.

Geophysical Research Letters



Figure 2. Seasonal cycle of (a) sea surface temperature (SST) averaged in Atl3. (b) *tau_x* averaged in WA4. (c) d20 in Atl3. (d–f) Same as (a)–(c) but for monthly standard deviation

2.2. Reference Data Set

In order to quantify the SST biases we use two different data sets ERSST v3b (Smith et al., 2008) and HadSST1.1 (Rayner et al., 2006). For wind stress, we use the ERA-Interim reanalysis (Dee et al., 2011). For the thermocline depth, as for the CMIP5 models, we use the 20° isotherm as a proxy computed from ORAS4 3-D temperature reanalysis monthly data Balmaseda et al. (2013). We use the recent period from 1979 to 2014, common to all reference data sets. For the climatological SST, Figure 1 compares results for 1979–2014 and 1870–1900 period.

2.3. Methods

We characterize four aspects of the Atlantic Niño:

- $\sigma_{JJ}SST$: the standard deviation of SSTs in the Atl3 box (3°S 3°N 20°W 0°E; Zebiak, 1993) in June–July (JJ), when the observed variability is at its maximum (Figure 2d).
- $Cor(tau_x, SST)$: the correlation between wind anomalies in May–June in WA4 and SST anomalies in Atl3 in JJ (Nnamchi et al., 2015). This index allows to quantify the wind-driven contribution, through the Bjerknes feedback, in SST variability.
- The position of the maximum of variability along the equator.

We use the difference between June and July SST as a proxy of the development of the cold tongue ($\Delta_{JJ}SST$ hereafter).

3. Results

Figures 1a–1c compares the multimodel (MM) mean with the observed climatological SST. The SST maximum is located too far to the south, and the east-west SST gradient along the equator is reversed. Therefore, a large warm bias exists along the equator and in the Angola-Benguela region. The bias is indeed exceeding the difference between the two observational data sets considered for the recent period. The bias is even larger when we compare the MM with the preindustrial period (Figures 1a and 1c). Figure 1d, which shows the observed standard deviation of JJ SST in the TA, illustrates the two main areas of interannual

Geophysical Research Letters



Figure 3. Each scatter plot relates two indicators of each CMIP5 models; for all figures black line shows the regression, number in the title the correlation marked with * (**) if it is significant at the 95% (99%) confidence interval; black, gray dots, and gray lines show the reference data sets. (a) Mean SST in JJ against the standard deviation of JJ SST ($\sigma_{JJ}SST$) in Atl3. (b) $\Delta_{JJ}SST$ against the $\sigma_{JJ}SST$ in Atl3. (c) Mean SST in JJ against the standard deviation of JJ SST ($\sigma_{JJ}SST$) in Atl3. (b) $\Delta_{JJ}SST$ against the $\sigma_{JJ}SST$ in Atl3. (c) Mean SST in JJ against $Cor(tau_x, SST)$. (d) $\Delta_{JJ}SST$ against $Cor(tau_x, SST)$. (e) Difference between April and June SST ($\Delta_{AJ}SST$) in Atl3 against the difference between April and May tau_x in WA4 ($\Delta_{AM}tau_x$). (f) $\Delta_{AM}tau_x$ against the difference between June and July d20 ($\Delta_{JJ}d20$) in Atl3. (g) $\Delta_{JJ}d20$ against the $\Delta_{JJ}SST$. (h) $\Delta_{AJ}SST$ against $\sigma_{JJ}SST$. (i) $\Delta_{AM}tau_x$ against $\sigma_{JJ}SST$. (j) $\Delta_{JJ}d20$ against $\sigma_{JJ}SST$. (k) Mean AJ SST against $\sigma_{JJ}SST$. (l) Mean AM tau_x against $\sigma_{JJ}SST$. (m) Mean JJ d20 against $\sigma_{JJ}SST$. SST = sea surface temperature; CMIP5 = Coupled Model Intercomparison Project Phase 5; JJ = June–July; AJ = April–June

variability in the TA: the equatorial Atlantic (Atl3) and in the Angola-Benguela region. In this study, we will focus on the Atl3 variability, associated with the Atlantic Niño, attributed to the Bjerknes feedback (Keenlyside & Latif, 2007). Despite a shift of the variability to the west, the error in MM interannual variability for Atl3 is smaller than the difference between the two observational products considered (Figure 1f). This suggests that models do not necessarily need to correctly represent the mean state to reproduce a relatively realistic interannual variability.

As introduced above, TA SST displays a very strong seasonal cycle, which is characterized by the development of a strong cold tongue between spring and summer. Figures 1g and 1h show that the seasonal cooling



Figure 4. Same as Figure 3 for (a) position of minimum JJ SST along the equator (meridional SST averaged between $3^{\circ}N$ and $3^{\circ}S$) against the position of maximum standard deviation of JJ SST (posmax $\sigma_{JJ}SST$). (b) Position of the strongest SST cooling along the equator between June and July ($\Delta_{JJ}SST$) against the position of maximum $\sigma_{JJ}SST$. (c) Averaged wind stress in May June in WA4 against the position of minimal d20 along the equator. (d) Position of minimal d20 along the equator against position of the strongest $\Delta_{JJ}SST$. SST = sea surface temperature; JJ = June–July

 $(\Delta_{JJ}SST, defined in section 2)$ develops in two regions: Atl3 and Angola-Benguela. Models fail in representing the latter but reasonably reproduce the cold tongue development in Atl3.

We examine the monthly evolution of climatology and interannual variability of the three variables involved in the Bjerknes feedback (Deppenmeier et al., 2016): SST over Atl3 in spring and summer, zonal wind stress over the Western equatorial Atlantic (WA4: 40°W2°W–4°S4°N), and d20 over Atl3 in summer for all the CMIP5 models and observations (Figure 2). For SST, the bias develops mainly between April and July, when the cold tongue in observations develops. The wind stress biases precede the SST bias; they are already strong in April–May. Winds (and d20) in the models accelerate (shallow) a few month later than in observations, consistently with Richter and Xie (2008) (Figures 2b and 2c). In terms of interannual variability, the model biases are more widespread than for climatology, except for d20 (Figures 2d–2f).

To estimate if there is a linkage between climatology and interannual variability errors, Figure 3a shows the regression between the strength of interannual variability in Atl3 and JJ SST climatology among the CMIP5 models. There is a weak relationship between the amplitude of the bias and the amplitude of the interannual variability ($\sigma_{JJ}SST$), which is not significant at 95% confidence level. However, if instead of considering the mean bias, we consider $\Delta_{JJ}SST$, as a proxy of the strength of the seasonal cold tongue development, a strong and significant (at 99% confidence level) relationship appears (Figure 3b). It is interesting to note that the MM correlation of $\Delta_{JJ}SST$ and SST_{JJ} is only -0.35 (not significant at the 90% confidence level). To properly simulate the Atlantic Niño, a correct cold tongue development in the TA between spring and summer is necessary. Figures 3c and 3d show that the wind-driven contribution of Atl3 variability, $Cor(tau_x, SST)$, is also strongly related with $\Delta_{JJ}SST$, which shows that the representation of the Bjerknes feedback in climatology is related to interannual variability in the Bjerknes feedback. A summary of how reliably each component in the Bjerknes feedback is simulated by the models follows (Figures 3e–3g)

- Step 1 (Figure 3e): Models simulating a stronger cooling between April and June ($\Delta_{AJ}SST$) are those simulating a stronger wind acceleration between May and June over WA4 ($\Delta_{MJ}tau_x$).
- Step 2 (Figure 3f): The models simulating stronger $\Delta_{MJ}tau_x$ are those simulating larger thermocline shallowing between June and July ($\Delta_{JJ}d20$). Here we consider the relative shallowing to the thermocline depth averaged over the same season. Individual models have different errors in thermocline depth (Figure 2), such that thermocline in models with very thin thermocline can hardly shoal as much as in models with deep thermocline.
- Step 3 (Figure 3g): Simulation of the relative thermocline shallowing is related to $\Delta_{JJ}SST$.

Figures 3h–3j show how the different components of the cold tongue formation are related to variability. It appears that there is a strong relationship between the strength of $\Delta_{MJ}tau_x$ and the Atlantic Niño. This highlights the importance of properly simulating the wind stress seasonal cycle to reliably simulate the Atlantic Niño, consistently with Richter et al. (2014, 2017).

Conversely, Figures 3k–3m show the same regressions as in Figures 3f–3h but using averaged values instead of difference between two months: Correlations are closed to 0, implying that the interannual variability is not directly linked to bias in climatology, but in the evolution of the seasonal cycle instead.

The variability in the models is not necessarily correctly situated in the Atl3 region. Figure 4 demonstrates how the positional error of the cold tongue and SST variability are related to each other. The position of maximum variability (pos max $\sigma_{JJ}SST$) is strongly correlated (0.75, significant at 99% confidence interval) with the position of the cold tongue development (Figure 4b), that is, the position of the maximum $\Delta_{JJ}SST$ along the equator (pos max $\Delta_{JJ}SST$). The position of the cold tongue maximum along the equator is in turn linked to the position of the d20 minimum (Figure 4d), itself is strongly related to the strength of wind stress in the WA4 region between May and June (Figure 4c).

4. Conclusion and Discussion

We have analyzed the relationship between climatology and interannual variability in the equatorial Atlantic using CMIP5 simulations. The representation of the Atlantic Niño is strongly linked to the cold tongue development and only weakly linked to the summer SST mean. In other words, the ability of a model to properly simulate the interannual variability in the TA is related to its ability to properly simulate Bjerknes feedback in climatology and the associated cold tongue development.

From this study, it appears that to improve the simulation of the Atlantic Niño, modelers should focus primarily on capturing the evolution of the seasonal cycle and the processes associated with it, rather than on representing climatological SST. We advocate for deeper attention, in the tuning exercise of coupled models, to the development of the seasonal cycle.

Investigating intermodel relationships does not allow us to fully understand the mechanisms connecting error in the seasonal cycle, summer SST climatology, and the variability. Using targeted sensitivity experiments with idealized wind field, ocean nudging, or anomaly coupling could be performed to further understand how wind phasing, thermocline depth, or mean SST at certain times of the year affect or not the variability. Beside, Deppenmeier et al. (2016) found that the relation between subsurface and SST anomalies was poorly reproduced by CMIP5 models, which is consistent with Figures 3g and 3j, so emphasis should also be put on better understanding the lack of connection between surface and subsurface anomalies.

Recently, Yang et al. (2017) and Nnamchi et al. (2015) have shown that some models are able to reproduce a realistic amplitude of variability driven by erroneous mechanisms. This might explain the outliers of the relationship in Figures 3 and 4, but it could also strongly affect the models' forecast skill in the region. More research is needed to quantify and understand sources of TA predictability in the seasonal forecasting context.

References

Bacha, M., Jeyid, M. A., Vantrepotte, V., Dessailly, D., & Amara, R. (2017). Environmental effects on the spatio-temporal patterns of abundance and distribution of Sardina pilchardus and sardinella off the Mauritanian coast (North-West Africa). Fisheries Oceanography, 26(3), 282–298. https://doi.org/10.1111/fog.12192

Balmaseda, M. A., Mogensen, K., & Weaver, A. T. (2013). Evaluation of the ECMWF ocean reanalysis system ORAS4. Quarterly Journal of the Royal Meteorological Society, 139(674), 1132–1161. https://doi.org/10.1002/qj.2063

Bjerknes, J. (1969). Atmospheric teleconnections from the equatorial Pacific ¹. *Monthly Weather Review*, 97(3), 163–172. https://doi.org/10.1175/1520-0493(1969)097<0163:ATFTEP>2.3.CO;2

Burls, N. J., Reason, C. J., Penven, P., & Philander, S. G. (2011). Similarities between the tropical Atlantic seasonal cycle and ENSO: An energetics perspective. *Journal of Geophysical Research*, *116*, C11010. https://doi.org/10.1029/2011JC007164

Carton, J. A., Cao, X., Giese, B. S., Da Silva, A. M., Carton, J. A., Cao, X., et al. (1996). Decadal and interannual SST variability in the Tropical Atlantic Ocean. *Journal of Physical Oceanography*, *26*(7), 1165–1175. https://doi.org/10.1175/1520-0485(1996)026<1165:DAISVI>2.0.CO;2
Cassou, C., Terray, L., & Phillips, A. (2005). Tropical Atlantic influence on European heat waves. *Journal of climate*, *18*, 2805–2811.

Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-interim reanalysis: configuration and performance of the data assimilation system. Quarterly Journal of the Royal Meteorological Society, 137(656), 553–597. https://doi.org/ 10.1002/qj.828

Acknowledgments

Chloe Prodhomme and Javier Garcia-Serrano were supported by the Spanish Juan de la Cierva (IJCI-2016-30802) and Ramon y Cajal (RYC-2016-21181) programs, respectively. The research has been partially funded by the EU-FP7 PREFACE Project (603521). CMP5 data are available from https://cmip.llnl.gov/ cmip5/index.html. Deppenmeier, A.-L., Haarsma, R. J., & Hazeleger, W. (2016). The Bjerknes Feedback in the tropical Atlantic. *Climate Dynamics*, 47, 2691–2707. https://doi.org/10.1007/s00382-016-2992-z

Ding, H., Keenlyside, N., Latif, M., Park, W., & Wahl, S. (2015). The impact of mean state errors on equatorial Atlantic interannual variability in a climate model. *Journal of Geophysical Research: Oceans*, 120, 1133–1151. https://doi.org/10.1002/2014JC010384

Dippe, T., Greatbatch, R. J., & Ding, H. (2017). On the relationship between Atlantic niño variability and ocean dynamics. Climate Dynamics, 51, 597–612. https://doi.org/10.1007/s00382-017-3943-z

Hourdin, F., Găuinusă-bogdan, A., Braconnot, P., Dufresne, J.-L., Traore, A.-K., & Rio, C. (2015). Air moisture control on ocean surface temperature, hidden key to the warm bias enigma. *Geophysical Research Letters*, 42, 10,885–10,893. https://doi.org/10.1002/2015GL066764

Keenlyside, N. S., & Latif, M. (2007). Understanding equatorial atlantic interannual variability. *Journal of Climate*, 20(1), 131–142. https://doi.org/10.1175/JCLI3992.1

Kucharski, F., Bracco, A., Yoo, J. H., & Molteni, F. (2008). Atlantic forced component of the Indian monsoon interannual variability. Geophysical Research Letters, 35, L04706. https://doi.org/10.1029/2007GL033037

Merle, J., Fieux, M., & Hisard, P. (2013). Annual signal and interannual anomalies of sea surface temperature in the eastern equatorial atlantic ocean. In Oceanography and Surface Layer Meteorology in the B/C Scale: Global Atmospheric Research Program Atlantic Tropical Experiment (Gate) (pp. 77–101). Pergamon. https://doi.org/10.1016/B978-1-4832-8366-1.50023-6

Nnamchi, H. C., Li, J., Kucharski, F., Kang, I.-S., Keenlyside, N. S., Chang, P., & Farneti, R. (2015). Thermodynamic controls of the Atlantic niño. *Nature Communications*, 6, 8895. https://doi.org/10.1038/ncomms9895

Okumura, Y., Xie, S.-P., Okumura, Y., & Xie, S.-P. (2004). Interaction of the Atlantic Equatorial Cold Tongue and the African Monsoon*. Journal of Climate, 17(18), 3589–3602. https://doi.org/10.1175/1520-0442(2004)017<3589:IOTAEC>2.0.CO;2

Rayner, N. A., Brohan, P., et al. (2006). Improved analyses of changes and uncertainties in sea surface temperature measured in situ since the mid-nineteenth century: The hadSST2 dataset. *Journal of Climate*, *19*(3), 446–469. https://doi.org/10.1175/JCLI3637.1

- Richter, I., & Xie, S.-P. (2008). On the origin of equatorial Atlantic biases in coupled general circulation models. *Climate Dynamics*, 31(5), 587-598. https://doi.org/10.1007/s00382-008-0364-z
- Richter, I., Xie, S. P., Behera, S. K., Doi, T., & Masumoto, Y. (2014). Equatorial Atlantic variability and its relation to mean state biases in CMIP5. Climate Dynamics, 42(1-2), 171–188. https://doi.org/10.1007/s00382-012-1624-5
- Richter, I., Xie, S.-P., Morioka, Y., Doi, T., Taguchi, B., & Behera, S. (2017). Phase locking of equatorial Atlantic variability through the seasonal migration of the ITCZ. *Climate Dynamics*, 48(11-12), 3615–3629. https://doi.org/10.1007/s00382-016-3289-y
- Rodríguez-Fonseca, B., Mohino, E., Mechoso, C. R., Caminade, C., Biasutti, M., Gaetani, M., et al. (2015). Variability and predictability of west african droughts: a review on the role of sea surface temperature anomalies. *Journal of Climate*, 28(10), 4034–4060. https://doi.org/10.1175/JCLI-D-14-00130.1
- Rodríguez-Fonseca, B., Polo, I., García-Serrano, J., Losada, T., Mohino, E., Mechoso, C. R., & Kucharski, F. (2009). Are Atlantic Niños enhancing Pacific ENSO events in recent decades? *Geophysical Research Letters*, *36*, L20705. https://doi.org/10.1029/2009GL040048
- Smith, T. M., Reynolds, R. W., Peterson, T. C., & Lawrimore, J. (2008). Improvements to NOAA's historical merged land-ocean surface temperature analysis (1880-2006). Journal of Climate, 21(10), 2283–2296. https://doi.org/10.1175/2007JCLI2100.1
- Taylor, K. E., Stouffer, R. J., Meehl, G. A., Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Terray, P., Masson, S., Prodhomme, C., Roxy, M. K., & Sooraj, K. P. (2015). Impacts of Indian and Atlantic oceans on ENSO in a comprehensive modeling framework. *Climate Dynamics*, 46(7), 2507–2533. https://doi.org/10.1007/s00382-015-2715-x

Voldoire, A., Claudon, M., Caniaux, G., Giordani, H., & Roehrig, R. (2014). Are atmospheric biases responsible for the tropical Atlantic SST biases in the CNRM-CM5 coupled model? *Climate Dynamics*, *43*(11), 2963–2984. https://doi.org/10.1007/s00382-013-2036-x

- Wang, C., Zhang, L., Lee, S.-k., Wu, L., & Mechoso, C. R. (2014). A global perspective on CMIP5 climate model biases. Nature Climate Change, 4(February), 201–205. https://doi.org/10.1038/NCLIMATE2118
- Yang, Y., Xie, S.-P., Wu, L., Kosaka, Y., Li, J., Yang, Y., et al. (2017). Causes of enhanced SST variability over the equatorial atlantic and its relationship to the atlantic zonal mode in CMIP5. *Journal of Climate*, 30(16), 6171–6182. https://doi.org/10.1175/JCLI-D-16-0866.1

Zebiak, S. E. (1993). Air-sea interaction in the equatorial Atlantic region, 6(8), 1567-1586.

https://doi.org/10.1175/1520-0442(1993)006<1567:AIITEA>2.0.CO;2